Author's response to reviews

Title: The Internet for weight control in an obese sample: results of a randomised controlled trial

Authors:

Aine McConnon (a.mcconnon@surrey.ac.uk)
Jennie E Cockroft (j.e.cockroft@leeds.ac.uk)
Emma L Harvey (e.l.harvey@leeds.ac.uk)
James D Thomas (j.d.thomas@leeds.ac.uk)
Joan K Ransley (j.k.ransley@leeds.ac.uk)
Darren C Greenwood (d.c.greenwood@leeds.ac.uk)
Laura Bojke (lg116@york.ac.uk)

Version: 5 Date: 27 November 2007

Author's response to reviews: see over
Responses to referee comments  (MS: 1250125268120734)

<table>
<thead>
<tr>
<th>Revisions</th>
<th></th>
</tr>
</thead>
<tbody>
<tr>
<td>Leave in Table II and add another table which is identical to Table I but includes only the responders.</td>
<td>A third table has been added which presents baseline data for RESPONDERS only split by intervention group (table II). Table II has become Table III.</td>
</tr>
<tr>
<td>Please document the type of consent obtained (verbal or written).</td>
<td>Amended on page 4. Written participant and GP consent was obtained.</td>
</tr>
<tr>
<td>Improve the abstract background to document both the purpose and the context of the study.</td>
<td>Additional text has been added to the abstract background so that it now documents the purpose and the context of the study.</td>
</tr>
</tbody>
</table>